Author's response to reviews

Title: Mycobacterium tuberculosis Complex Drug Resistance Pattern and Identification of Species Causing Tuberculosis in Two Regions of Cameroon

Authors:

Jean-Paul Assam-Assam (assamjean@yahoo.fr)
Veronique B Penlap (v.penlap@yahoo.fr)
Fidelis Cho-Ngwa (chongwa_ub@yahoo.co.uk)
Jean-Claude Tedom (tedomjc@yahoo.fr)
Irene Anie-Anyangwe (ianyangwe@yahoo.com)
Vincent P Titanji (vpk.titanji@yahoo.com)

Version: 2 Date: 8 November 2010

Author's response to reviews:

We added "in Two Regions of" in the title
We corrected a few grammatical errors in the manuscript
We included ethical clearance notes in the Materials and Methods section